Author’s response to reviews

Title: Usefulness of a stool to stabilize dental chairs for cardiopulmonary resuscitation (CPR)

Authors:

Norimasa Awata (smile_dc_osaka@yahoo.co.jp)
Takashi Hitosugi (hitosugi.takashi.724@m.kyushu-u.ac.jp)
Yoichiro Miki (miki.yoichiro.485@m.kyushu-u.ac.jp)
Masanori Tsukamoto (tsukamoto@dent.kyushu-u.ac.jp)
Yoshifumi Kawakubo (y.kawakubo.710@gmail.com)
Takeshi Yokoyama (yokoyama@dent.kyushu-u.ac.jp)

Version: 3 Date: 22 Jul 2019

Author’s response to reviews:

Dear Editor and Reviewers (BMC Emergency Medicine)

Thank you very much for reviewing our manuscript and offering valuable advice.

We have addressed Declarations which below. Especially, we noted an "Availability of data and materials" and revised the manuscript accordingly.

Declarations
- Ethics approval and consent to participate
- Consent to publish
- Availability of data and materials
- Competing interests
- Funding
- Authors' Contributions
- Acknowledgements
- Informed consent